Author's response to reviews

Title: Identification of New Biomarkers for Acute Respiratory Distress Syndrome by Expression-based Genome-wide Association Study

Authors:

Dmitry N Grigoryev (dgrigory@umaryland.edu)
Dilyara I Cheranova (dcheranova@cmh.edu)
Suman Chaudhary (schaudhary@cmh.edu)
Daniel P Heruth (dpheruth@cmh.edu)
Li Q Zhang (lqzhang@cmh.edu)
Shui Q Ye (sqye@cmh.edu)

Version: 5 Date: 3 July 2015

Author's response to reviews: see over
Dear BMC Pulmonary Medicine Editorial Team,

Please find attached our revised manuscript, which we would like to resubmit to BMC Pulmonary Medicine for publication. We have substituted term ALI with ARDS throughout the revised manuscript, according to the last definition provided by Berlin consensus. All substitutions are highlighted throughout the text.

Sincerely,

Dmitry N. Grigoryev, M.D., Ph.D.
Assistant Professor
Director, Omics Analysis Unit
Department of Pediatrics and Department of Biomedical and Health Informatics
Division of Experimental and Translational Genetics
Children's Mercy Hospitals and Clinics
University of Missouri-Kansas City School of Medicine
Pediatric Research Building, Rm# 4729.02
2401 Gillham Road
Kansas City, MO 64108
Email: dgrigoryev@cmh.edu
Phone: (816) 983-6500
Fax: (816) 983-6501